Author's response to reviews

Title: Voting with their feet - Predictors of Discharge Against Medical Advice in Aboriginal and Non-Aboriginal Ischaemic Heart Disease inpatients in Western Australia:

Authors:

Judith M Katzenellenbogen (judith.katzenellenbogen@uwa.edu.au)
Frank M Sanfilippo (frank.sanfilippo@uwa.edu.au)
Michael ST Hobbs (michael.hobbs@uwa.edu.au)
Matthew W Knuiman (matthew.knuiman@uwa.edu.au)
Dawn Bessarab (d.bessarab@curtin.edu.au)
Angela Durey (a.durey@curtin.edu.au)
Sandra C Thompson (sandra.thompson@uwa.edu.au)

Version: 5 Date: 11 August 2013

Author's response to reviews: see over
Dear Sir/Madam Editor

Voting with their feet - Predictors of Discharge Against Medical Advice in Aboriginal and Non-Aboriginal Ischaemic Heart Disease inpatients in Western Australia

We have revised the last point raised by the associate editor’s comments on our second revision.

1. Associate Editor's comments:

"Please report an $R^2$ value for each of the regression models".

We have calculated and reported on the $R^2$ values for all the models in Tables 2 and 3, currently highlighted in yellow.

I hope this final query has been adequately responded to and that our paper now meets your approval for publication.

Yours sincerely

Judith M Katzenellenbogen

First and Corresponding Author
Combined Universities Centre for Rural Health
University of Western Australia